Author's response to reviews

Title: Comparison of medicine availability measurements at health facilities: evidence from Service Provision Assessment surveys in five sub-Saharan African countries

Authors:

Yoonjoung Choi (ychoi@usaid.gov)
Paul Ametepi (Paul.Ametepi@icfi.com)

Version: 2 Date: 17 May 2013

Author's response to reviews:

Please see attached responses.